Author’s response to reviews

Title: How useful are registered birth statistics for health and social policy? A global systematic assessment of the availability and quality of birth registration data.

Authors:

David Phillips (davidp6@uw.edu)

Tim Adair (timothy.adair@unimelb.edu.au)

Alan Lopez (alan.lopez@unimelb.edu.au)

Version: 2 Date: 06 Dec 2018

Author’s response to reviews:

Blank